Author’s response to reviews

Title: Familial Mediterranean Fever, Inflammation and Nephrotic Syndrome: Fibrillary Glomerulopathy and the M680I Missense Mutation

Authors:

Dr Patrick W Fisher (pfisher002@md.northwestern.edu)
L. Tammy Ho (ltho@enh.org)
Robert Goldschmidt (rgoldschmidt@enh.org)
Ronald Semerdjian (resenerdijan@enh.org)
Gregory W Rutecki (grutecki@enh.org)

Version: 6 Date: 4 Aug 2003

I have enclosed the requested changes to our manuscript. Furthermore, I have outlined the changes point-by-point below. Additionally, the signed consent form has been faxed to the editorial office as requested.

REVISIONS TO OUR MANUSCRIPT:

1.) Pages 1 and 2 no corrections.
2.) Page 3, Background, 2 para, line 4 delete through its role as a transcription factor.
3.) Page 5, Case report, line 2 delete massive.
4.) Page 6, Discussion, We revised completely to address the requested changes from reviewer #2
5.) Conclusion, page 7: line 2, change remains to be determined to requires further elucidation.
Line 6: delete new
Line 7: change rise to increases.

Thank you again for your time and consideration of our manuscript for publication in your journal.

Sincerely,

Dr. Patrick W. Fisher
Email:pfisher002@md.northwestern.edu